Author's response to reviews

Title: The influence of geographic and climate factors on the timing of dengue epidemics in Peru, 1994-2008

Authors:

Gerardo Chowell (gchowell@asu.edu)
Bernard Cazelles (cazelles@biologie.ens.fr)
Helene Broutin (broutinh@mail.nih.gov)
Cesar V Munayco (cvmunayco@dge.gob.pe)

Version: 4 Date: 20 May 2011

Author's response to reviews: see over
Submission of a revised research article

Tempe, May 20, 2011

Dear Editor,

Please find enclosed our revised manuscript, “The influence of geographic and climate factors on the timing of dengue epidemics in Perú, 1994-2008”, by Gerardo Chowell, Bernard Cazelles, Helene Broutin, Cesar V. Munayco, for consideration for publication in *BMC Infectious Diseases*.

This paper has not been submitted elsewhere, and all authors have approved the final version of the manuscript.

Thank you very much.

Sincerely,

Gerardo Chowell

Corresponding author:
Gerardo Chowell, PhD
Assistant Professor
School of Human Evolution and Social Change
Arizona State University
And
Fogarty International Center, NIH

Box 872402
Tempe, AZ 85287
Phone: 480-965-4730
Fax: 480-965-7671
Email: gchowell@asu.edu